Author’s response to reviews

Title: Open globe and penetrating eyelid injuries from fish hooks

Authors:

Konstantine Purtskhvanidze (konstantine.purtskhvanidze@uksh.de)
Mark Saeger (mark.saeger@uksh.de)
Felix Treumer (felix.treumer@uksh.de)
Bernhard Nölle (bernhard.noelle@uksh.de)
Johann Roider (johann.roider@uksh.de)

Version: 2 Date: 10 Dec 2018

Author’s response to reviews:

Author’s response to reviews

Title: Open globe and penetrating eyelid injuries from fish hooks

Authors:

Konstantine Purtskhvanidze (konstantine.purtskhvanidze@uksh.de)
Mark Saeger (mark.saeger@uksh.de)
Felix Treumer (felix.treumer@uksh.de)
Bernhard Nölle (bernhard.noelle@uksh.de)
Johann Roider (johann.roider@uksh.de)

Version: 3, 8. November 2018
1. We note that you have included the following statement in the Consent for Publication section of your manuscript: Nevertheless, written informed consent of the patients to participate in the research was obtained”.

According to our editorial policies, "For all research involving human subjects, informed consent to participate in the study should be obtained from participants (or their parent or legal guardian in the case of children under 16)".

Furthermore, in regards to consent for publication, "For all manuscripts that include details, images, or videos relating to an individual person, written informed consent for the publication of these details must be obtained from that person (or their parent or legal guardian in the case of children under 18)".

As some of the patients included in your study are under the age of 16, please clarify whether consent to participate was obtained from the parents/legal guardians of these patients.

In addition, please clarify whether consent for publications was obtained from the patients, or the parents/legal guardians of the patients under the age of 18.

Please ensure that a clear statement on consent to participate is included on the Ethics Approval and Consent to Participate section of your manuscript. Please also ensure that a clear statement on consent for publication is included in the Consent for Publication section of your manuscript.

We added this information (line 231; line 235).

2. Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

We included this statement (line 246).

3. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

We added a list of all abbreviations (line 206).
4. Please remove the Authorship form from the file inventory.

We removed this form.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.